Dear Editor,

I hereby submit you a revised version of the article entitled "Acute presentation of vasospastic angina induced by oral capecitabine: a case report".

(MS: 1139669541007837)

All reviewer (s)’ comments were taken into account and point to point answers are included at the bottom of this letter.

Please do not hesitate to contact me if necessary,

Thank you in advance,

Sincerely yours,

Prof. K.A. Charalabopoulos MD, PhD

Department of Physiology, Clinical Unit, Medical Faculty, Democritus University of Thrace

Panepistimioupoli, Dragana

68100, Alexandroupolis, Greece

Tel: 003 25510-30538
POINT TO POINT CHANGES

IN GENERAL
All reviewer (s)' suggestions were taken into account.

IN PARTICULAR
1. The Ethnicity of the patient in the Abstract and Case presentation sections was included (“of Greek ethnicity”).
2. Keywords representing the main content of the article were added after the Abstract section.
3. The ‘Case Report’ section header was changed to ‘Case Presentation’. The description of the patient’s gender from ‘woman’ to ‘female’ was changed.
4. An acknowledgement section at the end of the manuscript before the reference list was added.
5. The source(s) of funding for all authors was declared.
6. The figure files were edited as they should be closely cropped.